Author's response to reviews

Title: A cross sectional study: Latrine coverage and associated factors among rural communities in Bahir Dar Zuria, Ethiopia.

Authors:

Worku A Yalew (workuawo@yahoo.com)
Semahegn M Bayabil (semahegnmuche@yahoo.com)

Version: 3 Date: 10 September 2012

Author's response to reviews:

Dear Editorial Office,

Subject: Submission of manuscript

Manuscript which we sent previously; entitled with “Latrine coverage and associated factors among rural communities in Bahir Dar Zuria District, West Gojjam Zone, Amhara Region, Ethiopia. 2012” was edited and modified as “A cross sectional study: Latrine coverage and associated factors among rural communities in Bahir Dar Zuria, Ethiopia.” and also the abstract and the main body of the document was edited by language editors.

I sent the manuscript to be reviewed and considered for publication to your journal. The manuscript is more suitable to the scope your journal. In doing so, we would like to confirm you that, the manuscript is original, not sent elsewhere for publication and the copyright will be under your journal.

Thank you for your consideration of our manuscript

Worku Awoke Yalew & Semahegn Muche Bayabil